Author's response to reviews

Title: Relationship between participation in leisure activities and constraints on Taiwanese breastfeeding mothers during leisure activities

Authors:

Hsueh-wen Chow (hzc110@gmail.com)
Yin-Han Dong (rb6981025@gmail.com)

Version: 2 Date: 6 September 2012

Author's response to reviews: see over